Author’s response to reviews

Title: Coping with ill-health: Health care facility, chemist or medicinal plants? Health-seeking behaviour in a Kenyan wetland

Authors:

Carmen Anthonj (carmen.anthonj@gmail.com)

Peter Giovannini (peter.giovannini@gmail.com)

Thomas Kistemann (thomas.kistemann@ukbonn.de)

Version: 4 Date: 28 Apr 2019

Author’s response to reviews:

Dear Editors,

we tried our best to minimize overlaps between our manuscript and the PhD thesis that this paper is based on (Anthonj C (2017) Mosquitoes and Maladies among Men using Marshes. Is the Contraction of Diseases in Wetlands a Question of Use? Behaviours, Risk Assessments and Perceptions from the Ewaso Narok Swamp, Kenya. Doctoral thesis, Bonn, 267 pages) the maximum way possible. We rephrased again major parts of the results, discussion and conclusions sections as per your feedback.

Thank you very much for taking the time to help us improve this manuscript.

Sincerely

the Authors